Author’s response to reviews

Title: PREDICTORS OF LOSS TO FOLLOW-UP AMONG CHILDREN ON LONG-TERM
ANTI-RETROVIRAL THERAPY IN ZAMBIA (2003-2015)

Authors:

Jane Mutanga (janemutanga@gmail.com)
Simon Mutembo (simon.mutembo@gmail.com)
Amara Ezeamama (Amara.Ezeamama@hc.msu.edu)
Xiao Song (xsong.uga@gmail.com)
Robert Fubisha (fubisha@yahoo.co.uk)
Kunda Mutesu-Kapembwa (mutesukunda@yahoo.com)
Derrick Sialondwe (dsialondwe@yahoo.co.uk)
Brendah Simuchembu (brendahsimuchembu@yahoo.com)
Jelita Chinyonga (jelitachinyonga@gmail.com)
Philip Thuma (phil.thuma@macharesearch.org)
Christopher Whalen (ccwhalen@uga.edu)

Version: 2 Date: 24 Jul 2019

Author’s response to reviews:

22nd July 2019,

Dear Editor,

Thank you for your comments.

We have cited Figure 2 and have attached a clean version of the manuscript.

Sincerely,

Jane N. Mutanga